Cover Page for Homework 3 Problem 8

University of Toronto CSC364 Computer Science Summer 2004

Student Number:

Name:

Tutorial: Paul Matei Hamed Jon Victor (Circle one) A - D E - Ka Ke - Lo Lu - S T - Z

You must acknowledge any references (books, webpages, people, etc.) which you used to assist you in solving the problem as well as any student which you assisted. Failure to do so will be considered an academic offense. You do not need to list the course textbook or lecture notes. Write “NONE” if you did not consult any references or discuss the problem with anyone.

Collaboration note:

By signing your name you indicate that the above list of references and people is correct and that you have read and understood the course policy on academic honesty.

Signature: